Author's response to reviews

Title: The prediction of treatment outcomes by early maladaptive schemas and schema modes in obsessive-compulsive disorder

Authors:

Nicola Thiel (nicola.thiel@uniklinik-freiburg.de)
Brunna Tuschen-Caffier (tuschen@psychologie.uni-freiburg.de)
Nirmal Herbst (nirmal.herbst@uniklinik-freiburg.de)
Anne Katrin Kuelz (anne.katrin.kuelz@uniklinik-freiburg.de)
Christoph Nissen (christoph.nissen@uniklinik-freiburg.de)
Elisabeth Hertenstein (Elisabeth.hertenstein@uniklinik-freiburg.de)
Ellen Gross (ellen.gross@uniklinik-freiburg.de)
Ulrich Voderholzer (UVoderholzer@schoen-kliniken.de)

Version: 4 Date: 24 November 2014

Author's response to reviews: see over
Dear Dr. Gwyneth Zai and Mr. Carlo Rye Chua,

We would like to resubmit our revised version of the manuscript ‘The prediction of treatment outcomes by early maladaptive schemas and schema modes in Obsessive-Compulsive Disorder’ for publication in BMC Psychiatry.

Please find enclosed our point-by-point answer to the comment and the revised version of our manuscript.

We believe that our manuscript has further improved. Thank you very much for considering our work for publication.

Sincerely,

Nicola Thiel
Reply to comment:

We would like to thank Dr. Gwyneth Zai and Mr. Carlo Rye Chua for the review of our manuscript and the comment.

Reply to comment of the Editor:

1.) I would like a clarification from the authors regarding the definition of age of onset (first onset of symptom, first diagnosis, or first hospitalization).

We apologize that we did not mention the information before. We thank the Editor for this important recommendation. Following the advice, we now provide the information on page 10, line 269.